Author's response to reviews

Title: Mutations in WDR62 gene in Pakistani families with autosomal recessive primary microcephaly

Authors:

Rizwana Kousar (rizwana_aiou@yahoo.com)
Muhammad Jawad Hassan (hassanraja12@hotmail.com)
Bashra Khan (bushrakhan1900@yahoo.com)
Sulman Basit (basitphd@bs.qau.edu.pk)
Saqib Mahmood (sgb_medgen@yahoo.com)
Asif Mir (mir77uspk@yahoo.com)
Wasim Ahmad (wahmad@qau.edu.pk)
Muhammad Ansar (ansar@qau.edu.pk)

Version: 3 Date: 13 September 2011

Author's response to reviews: see over
September 13, 2011

Editor
BMC Neurology

Dear Sir,

Here we are submitting our revised manuscript entitled, “Mutations in WDR62 gene in Pakistani families with autosomal recessive primary microcephaly” by Kousar et al. as research article for publication in BMC Neurology.

We have incorporated all the suggestions of Associate Editor mentioned in the email to improve the manuscript. Thank you for your time and efforts to review our manuscript for publication in BMC Neurology.

Sincerely,

Muhammad Ansar, PhD
Assistant Professor
Department of Biochemistry
Faculty of Biological Sciences
Quaid-i-Azam University, Islamabad
PAKISTAN
Tel. No. 92-51-90643146
E.mail: ansar@qau.edu.pk
mansar_76@yahoo.com